Author’s response to reviews

Title: The Ethics of Caring for Hospital-Dependent Patients

Authors:

Calvin Sung (calvin.sung@quinnipiac.edu)

Jennifer L Herbst (jennifer.herbst@quinnipiac.edu)

Version: 2 Date: 29 Nov 2017

Author’s response to reviews:

Editor Comments: 1 Please include the aims as short sentence in the Background section of the Abstract. Please do not put it is a separate section, just a single sentence at the end of background which begins something like ‘The aim of this paper is to….’ A sentence about the aim of the paper has been included at the end of the Background section of the Abstract (Lines 14-15). 2 The cases of John Smith and Jane Doe can you clarify, in text, if these are hypothetical examples or real cases. If this is a real case, it contains potentially identifiable information. It is BMC policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper (http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9). Currently, the age and gender information, in combination with other identifiers, may compromise patient/participant anonymity. For all manuscripts that include details relating to an individual person, WRITTEN informed consent for the publication of these details must be obtained from that person (or their parent or legal guardian in the case of children under 18). Please provide the Consent for Publication as the following format if you have obtained the Consent for Publication in written form. “Written informed consent was obtained from the participants for publication of this Article and any accompanying tables/images. A copy of the written consent is available for review by the Editor of this journal.” The subjects in these cases have been clearly noted as hypothetical (Lines 92 & 119). 3 Please provide the specific grant code for the funding if there is any in Funding A grant identification number (2142001) has been included. 4 Upon resubmission, Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published. If you wish to respond to these revision requests, please insert the information in to the designated input box only All tracked changes have been removed and a clean copy of the manuscript is being submitted in its final form. No other supplemental files are being submitted.